Author’s response to reviews

Title: Effect of Hyperplasia and Apoptosis in Neural Stem Cells Treated with Sevoflurane

Authors:

Jianlei Qiu (qiujianlei2012@163.com)
Pengcai Shi (qiujianlei2012@163.com)
Wude Mao (qiujianlei2012@163.com)
Yuyi Zhao (qiujianlei2012@163.com)
Wenshuai Liu (qiujianlei2012@163.com)
Yuelan Wang (wyldgf@yeah.net)

Version: 2 Date: 23 September 2014

Author’s response to reviews:

We are very grateful for the editor’s assessment of our work. We very much sympathize with the opinions expressed above. In the revised version, we have added line and page numbers and listed all abbreviations ahead of the competing interests and authors’ contributions. In response to the valuable remarks expressed above, we have made corrections. We hope these changes will be positively received by editor.